Author's response to reviews

Title: The effectiveness of extracorporeal shock wave therapy for the treatment of lower limb ulceration: a systematic review

Authors:

paul butterworth (paul.butterworth@scu.edu.au)
tom walsh (tomwalsh_pod@me.com)
yvonne pennisi (yvonne.pennisi@scu.edu.au)
Anna Du Chesne (anna.duchesne@scu.edu.au)
Christoph Schmitz (christoph_schmitz@med.uni-muenchen.de)
susan nancarrow (susan.nancarrow@scu.edu.au)

Version: 2 Date: 6 October 2014

Author's response to reviews:

October 6th, 2014
Dear Editors,

Please find enclosed our revised manuscript titled:

The effectiveness of extracorporeal shock wave therapy for the treatment of lower limb ulceration: a systematic review, for consideration for publication in Journal of Foot and Ankle Research.

Chronic ulceration is a public health issue and is dealt with on a regular basis by your readership. We believe this paper has been revised to meet all of the reviewers’ comments where possible.

Each of the authors has read and concurs with the content in the final manuscript. The material within has not been, and will not be, submitted for publication elsewhere. We believe that our study is novel and important. Accordingly, we hope that our manuscript may be of interest to your journal.

Yours sincerely,

Paul Butterworth
Primary author